Author's response to reviews

Title: Evaluating the impact of depression, anxiety & autonomic function on health related quality of life, vocational functioning and health care utilisation in acute coronary syndrome patients: the ADVENT study protocol

Authors:

John C Oldroyd (john.oldroyd@monash.edu)
Sheila Cyril (sheila.cyril@monash.edu)
Bhanuja Wijayatilaka (bhanuja.wijayatilaka@monash.edu)
Adrienne O&8217;Neil (AONEIL@barwonhealth.org.au)
Dean McKenzie (dean.mckenzie@monash.edu)
Silva Zavarsek (silva.zavarsek@monash.edu)
Kristy Sanderson (kristy.sanderson@utas.edu.au)
David Hare (david.hare@unimelb.edu.au)
Aaron Fisher (afisher@berkeley.edu)
Andrew Forbes (andrew.forbes@monash.edu)
Craig Barr-Taylor (btaylor@stanford.edu)
David Clarke (David.Clarke@monash.edu)
Ian Meredith (ian.meredith@myheart.id.au)
Brian Oldenburg (brian.oldenburg@monash.edu)

Version: 2

Date: 28 October 2013

Author's response to reviews:

Dr John Oldroyd
Research Fellow
BMC Cardiovascular Disorders
28th October 2013

Paper Title:
MS: 1411351502111390 Evaluating the impact of depression, anxiety & autonomic function on health related quality of life, vocational functioning and health care utilisation in acute coronary syndrome patients: the ADVENT study protocol

Dear Editor

Please find attached a copy of our updated protocol paper for the ADVENT study as per your email of 26th October 2013.

We have made the following changes:
1) Inserted the Figure after the references

Also, we can confirm that we have ethical approval and external funding for our
study. We received ethical approval from Monash Health (formerly known as Southern Health) where the study will be conducted. We received external funding from a government organisation the National Health and Medical Research Council, Australia. We attach evidence of our approvals including the amount of funding received.

There are no other manuscripts that are under submission with other journals based on this study protocol.

Yours sincerely,

Dr John Oldroyd
Corresponding author